Author’s response to reviews

Title: Heterogeneous global health stock and growth: Quantitative evidence from 140 countries, 1990-2100

Authors:

Isma Addi Jumbri (isma@utem.edu.my)

Shinya Ikeda (shinyaik@gmail.com)

Shunsuke Managi (managi.s@gmail.com)

Version: 1  Date: 17 Oct 2018

Author’s response to reviews:

We received a following comment from an editor: This manuscript has some potential interest but is far too long. The manuscript must be reduced by at least 40% before being sent to peer-review. Now we reduced the manuscript by 45%.